Author's response to reviews

Title: Parental reports of the oral health-related quality of life of children with cerebral palsy

Authors:

Jenny Abanto (jennyaa@usp.br)
Thiago S Carvalho (thiagosaads@usp.br)
Marcelo S Bönecker (bonecker@usp.br)
Adriana O.L. Ortega (aliraort@uol.com.br)
Ana L Ciamponi (analidia@usp.br)
Daniela P Raggio (danielar@usp.br)

Version: 6 Date: 21 May 2012

Author's response to reviews: see over
Cover letter

Dear Editor,

Thank you for your attention on considering this manuscript to be published at BMC. All the suggested considerations were addressed in this revised manuscript, as will be described as follow:

- The author’s contributions and acknowledgements were included.
- The revised manuscript conforms to the journal style.

Thank you for your attention,

Ana Lidia Ciamponi